Author's response to reviews

Title: The initial experience of reduced port surgery using a 2-surgeon technique for colorectal cancer

Authors:

Jo Tashiro (tashiro@saitama-med.ac.jp)
Shigeki Yamaguchi (s_yama@saitama-med.ac.jp)
Toshimasa Ishii (ishii@saitama-med.ac.jp)
Hiroka Kondo (hirokak@saitama-med.ac.jp)
Kiyoka Hara (k_hara@saitama-med.ac.jp)
Ryuichi Kuwahara (geno25@saitama-med.ac.jp)

Version: 4
Date: 7 July 2015

Author's response to reviews:

July 7, 2015
Editor-in-Chief Dr. Kostiantyn Kopchak
Mr. Eden Nonato
BMC Surgery
Dear Editor:

1. Please directly include the tables at the end of the manuscript and remove the table files from the additional files.
ANS: included

2. We recommend that you copyedit the paper to improve the style of written English.
ANS: brushed up

Sincerely,
Jo Tashiro
Yamane, Hidaka-shi, Saitama 350-1298, Japan 350-1298
Phone No: +8142-984-4111
Fax No: +8142-984-0054
Email Address: tashiro@saitama-med.ac.jp